

### Non-steroidal anti-inflammatory drugs in the pharmacological management of osteoarthritis in the very old: prescribe or proscribe?

Christian Cadet, Emmanuel Maheu

### ▶ To cite this version:

Christian Cadet, Emmanuel Maheu. Non-steroidal anti-inflammatory drugs in the pharmacological management of osteoarthritis in the very old: prescribe or proscribe?. Therapeutic advances in musculoskeletal disease, 2021, 13, pp.1759720X2110221. 10.1177/1759720x211022149. hal-03278590

### HAL Id: hal-03278590 https://hal.sorbonne-universite.fr/hal-03278590v1

Submitted on 5 Jul2021

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

Ther Adv Musculoskel Dis

Article reuse guidelines: sagepub.com/journals-

2021 Vol 13-1-13

DOI: 10.1177/ 1759720X211022149 © The Author(s), 2021.

permissions

Review

Therapeutic Advances in Musculoskeletal Disease

# Non-steroidal anti-inflammatory drugs in the pharmacological management of osteoarthritis in the very old: prescribe or proscribe?

# Christian Cadet<sup>(D)</sup>, Emmanuel Maheu and The French AGRHUM Group (Association Geriatric and RHeUMatology)

Abstract: Osteoarthritis (OA) is the most common form of arthritis worldwide, and ranges in the top 5–10 most disabling diseases. Contrary to common opinion, this disease is severe, often symptomatic, and may lead to loss of mobility and independence, as well as being responsible for increased frailty and excess mortality [standardized ratio: 1.55 (95% confidence interval, CI: 1.41–1.70]]. The incidence of OA increases dramatically with age in an increasingly ageing world. Therefore, practitioners involved in the management of OA often have to manage very old patients, aged 75-80 years and above, as part of their daily practice. Treatment options are limited. In addition to education and physical treatments, which are at the forefront of all treatment recommendations but require a low level of symptoms to be implemented, many pharmacological options are proposed. Nonsteroidal anti-inflammatory drugs (NSAIDs) can be used as a second-line treatment but with great caution. However, the precise incidence of cardiovascular, renal, and gastrointestinal adverse events in very elderly patients is unclear. All of these risks are increased in the elderly. The relative risks can be extrapolated from various studies. However, what is the absolute risk according to age categorization? The answer to this guestion is important because NSAIDs should be used in very elderly patients with OA only if full information has been provided and the decision to prescribe this treatment is shared between the patient and their doctor. This article reviews the risks and currently available recommendations, and proposes practical options and warnings to allow for a responsible and limited use of NSAIDs in the very old.

### Plain language summary

### NSAIDS in the very Old : Prescribe or Proscribe?

• Osteoarthritis (OA) in the very old is a serious disease leading to loss of independence, frailty, and excess mortality. Quantitative data from clinical trials and population-based observational studies on the risk of NSAID-related side effects allow the prescriber to provide more accurate information to each patient. If there is no contraindication, the decision to initiate NSAID therapy in a very old OA patient should be made in a shared manner, with the patient fully informed of the risks.

Keywords: NSAIDs, osteoarthritis, very old, benefit/risk, shared prescription

Received: 17 January 2021; revised manuscript accepted: 15 May 2021.

Correspondence to: Christian Cadet Medical Practice, 4, Place Martin Nadaud, PARIS, 75020, France chcadet@sfr.fr

**Emmanuel Maheu** Hospital Saint-Antoine, Paris, Île-de-France, France

journals.sagepub.com/home/tab



Creative Commons Non Commercial CC BY-NC: This article is distributed under the terms of the Creative Commons Attribution-NonCommercial 4.0 License (https://creativecommons.org/licenses/by-nc/4.0/) which permits non-commercial use, reproduction and distribution of the work without further permission provided the original work is attributed as specified on the SAGE and Open Access pages (https://us.sagepub.com/en-us/nam/open-access-at-sage).

### Introduction

Treating multiple osteoarthritis (OA) pain in very old patients ( $\geq$ 75–80 years) is a frequent challenge for general practitioners, rheumatologists, and orthopedists in daily clinical practice. The objective to relieve pain in such frail people must be weighed against iatrogenic risks.

Treatment options are limited. International recommendations on OA management rightly advocate a non-pharmacological approach.<sup>1–6</sup> However, these treatments are often slow-acting, insufficiently effective, and difficult to implement in very old patients suffering from severe pain.<sup>7</sup> Reducing pain, bringing patients to the Patient Acceptable Symptom State,<sup>8</sup> seems a mandatory prerequisite for any physical treatment.

Because of its good safety record, paracetamol is used widely but its effectiveness on OA pain is currently questioned. A recent meta-analysis suggested that paracetamol was almost useless.9 Non-steroidal anti-inflammatory drugs (NSAIDs) are more likely to reduce pain.<sup>10</sup> With NSAIDs, the number needed to treat to achieve a reduction of at least 50% in pain intensity after 12 weeks of treatment is between 4.7 and 8.4, depending on the drug used (naproxen 1000 mg, celecoxib 200 mg or ibuprofen 2400 mg).<sup>11</sup> In knee OA, a recent systematic review and meta-analysis found statistically significant and clinically important effects of NSAIDs on pain as early as 2 weeks after initiation with a standardized mean difference (SMD) of -0.43 [95% confidence interval (CI) -0.48, -0.38]. This treatment effect remained statistically significant up to 26 weeks [SMD -0.21 (95% CI -0.39, -0.03)].<sup>12</sup>

Opioids, recommended as a second-line treatment, are effective but very badly tolerated. Although there is no specific accurate data on safety in the elderly, a poor safety profile has been outlined, in particular with an increased risk of falls and related fractures, of confusion, renal impairment, and all-cause mortality, compared with NSAIDs.<sup>13</sup> The global number needed to harm has been estimated as high as five.<sup>14</sup>

The use of systemic corticosteroids in the treatment of pain in osteoarthritic patients has been the subject of only a few publications with contradictory results.<sup>15,16</sup> Because of their potential side effects [water retention, diabetes destabilization, osteoporosis, adverse cardiovascular (CV) effects...], their prescription in elderly people with OA is not recommended; they could be used for very short-term or palliative administration in patients at the end of life.<sup>5</sup>

Despite their frequent and sometimes serious side effects, should we still use NSAIDs in very old people? Should we discourage our patients from continuing to use over-the-counter (OTC) NSAIDs? And can we establish and quantify the risks of NSAIDs in this segment of the population and accurately inform the patient?

This is crucial for true shared decision-making. Patients are more willing than their physicians to accept a risk of harm for successful NSAID therapy.<sup>11</sup> The purpose of this article is to examine the potential risks and benefits of NSAID therapy in very elderly patients with OA and to provide an update on data that may help share the decision to prescribe NSAIDs with the patient.

### Methods

We first searched the Medscape and Cochrane databases using the search terms "older" or "very old" or "elderly" or "geriatric" and "osteoarthritis" and "NSAID" or "non-steroidal anti-inflammatory drugs."

After selecting appropriate publications, we searched their bibliographies manually for relevant papers, then extended the search to metaanalyses, recommendations, and reviews of recent literature dealing with the treatment of OA and pain in geriatrics.

### Who are we talking about?

This viewpoint focuses on patients aged over 75, even 80, years old in whom OA is extremely frequent. Its related functional disability combined with other co-morbidities, is responsible for a high burden in terms of pain, functional impairment, and loss of independence. In this age group, one person in five with hip or knee OA is inadequately relieved by analgesics and still experiences severe or extreme pain.<sup>17</sup>

Ageing, with multiple co-morbidities such as impaired renal, cardiovascular, and hepatic functions and changes in body composition, modifies drug pharmacokinetics and pharmacodynamics, increasing their iatrogenicity.

### What are we talking about?

OA is a severe condition that is not sufficiently often taken into consideration. In a populationbased cohort, with a 14-year median follow up, OA patients had a standardized mortality ratio of 1.55 (95% CI 1.41-1.70) compared with the general population.<sup>18</sup> In a population cohort aged 55+ years with at least moderately severe symptomatic hip and/or knee OA, the adjusted hazardratio (aHR) of all-cause mortality per unit increase of the Health Assessment Questionnaire (HAQ) walking score was 1.30 (95% CI 1.22-1.39, p < 0.001); the aHR when using a walking aid versus not using one was 1.51 (95% CI 1.34-1.70, p < 0.0013).<sup>19</sup> More recently, in a sample of 1709 participants in the Johnston County Osteoarthritis Project (mean age of participants  $59.5 \pm 9.5$  years) with symptomatic knee OA and not reporting CV disease (CVD) at baseline, it was shown that functional worsening, assessed by the Health Assessment Questionnaire (HAQ), the time to complete five chair stands, and gait speed during an 8-foot walk was associated with a significantly increased risk of CVD on all three functional measures: HAO odds ratio (OR) = 2.49 (95% CI 1.90-3.25), chair stands OR=1.58 (95% CI 1.20-2.08), and 8-foot walk OR=1.53 (95%CI 1.15-2.04).20

In a European survey of 2455 individuals aged 65-85 years, clinical OA at any site (hip, knee, hand) was associated with frailty.<sup>21</sup> Increasing walking capacity and reducing disability with effective treatments probably reduce OA-associated mortality risk. Most patients with OA are willing to accept some additional risk of stomach bleeding and/or heart attack/stroke to gain pain relief .<sup>22</sup> Statistical studies support a bimodal distribution of pain treatment response in chronic pain patients.<sup>23</sup> Only a few OA patients are good responders with a decrease in pain of more than 50%.<sup>11</sup> It is in this group of patients that the benefit/risk ratio of NSAID treatment could be the most valuable. However, although the prevalence of OA increases dramatically with age, we lack specific data in very old OA patients.

### What are specific NSAID-related risks?

The side effects of NSAIDs are numerous and can be serious: hypersensitivity, dizziness and falls, headaches, rare hepatotoxicity, drug interactions, possible chondrotoxicity, etc. The major side effects of NSAIDs are gastrointestinal GI complications, renal disturbances, and CV events. These side effects are related to the inhibition of cyclo-oxygenase (COX) enzyme activity and prostaglandin synthesis. They can be severe, leading to death, especially in frail patients. They may occur early in the course of treatment, although in most studies the risks appear to increase with longer use or higher doses.

*CV adverse effects.* The adverse effects of NSAIDs on the CV system appear to be based primarily on two distinct mechanisms:

- First, water and salt retention, dependent primarily on the inhibition of renal medullary COX-2, leading to increased renin production and the possible development of reno-vascular hypertension and heart failure. The activity of COX-2-derived prostanoids in renal medullary occurs mainly under conditions of sodium depletion or overloading. This effect, which is not very large in healthy people, can have important repercussions when there are, as is common in elderly patients, pre-existing pathological conditions such as high blood pressure, liver or kidney failure, etc.
- Secondly, an effect on the vascular endothelium and platelets leading to a thrombotic risk. This action was first attributed to an imbalance in the production of thromboxane A2 (TXA2), which is pro-aggregating and vasoconstrictive and primarily under the control of COX-1, and the COX-2 related production of prostacyclin (PGI2), which is anti-aggregating, vasodilating, and inhibits the vascular response to TXA2. This hypothesis would explain the better CV safety of naproxen, which strongly inhibits COX-1 with an aspirin-like effect that counterbalances the negative effect of COX-2 inhibition. According to this hypothesis, the most COX-2 selective NSAIDs should present the greatest thrombotic risk, which has not been verified for celecoxib. It has now been established that COX-2 selectivity alone is not the cause of the CV risks of non-steroidal anti-inflammatory drugs,<sup>24</sup> and each NSAID may have a unique safety profile. Some authors have postulated that CV risks are much more related to the potency of COX-2 inhibition than to the selectivity of the drug; the most effective drugs would also be the most hazardous to the CV system.25 This issue remains controversial.26

The nature and frequency of CV side effects of NSAIDs remain debated, with significant differences and sometimes conflicting results between randomized, double-blind clinical trials conducted in selected populations, excluding patients at higher risk of side effects, and epidemiological and case-control studies that are more subject to statistical biases, in particular protopathic biases.<sup>25,27</sup>

Hypertension. Inhibition of COX-2 blocks its natriuretic effect via the renin-angiotensin pathway. NSAIDs also inhibit the vasodilating effect of prostaglandin and stimulate the production of various vasoconstrictor factors such as endothelin-1. In the general population,<sup>28</sup> as in the elderly population,<sup>29</sup> NSAID therapy increases systolic blood pressure by approximately 5mmHg on average and has little or no effect on diastolic blood pressure. This increase would be different with different NSAIDs and greater with higher doses of NSAIDs.<sup>30</sup> It is particularly marked with the prior use of antihypertensive drugs, even if the blood pressure was previously effectively treated,<sup>28</sup> mainly diuretics, especially loop diuretics, angiotensin-converting enzyme inhibitors.<sup>31</sup> This situation is very common in the elderly; in France, in 2014, 58% of subjects over the age of 75 were receiving antihypertensive treatment, 39% bitherapy, 14% triple therapy, and 7% four or more different treatments.32

*Heart failure.* The risk of heart failure (HF) is related directly to elevated blood pressure, with increased systemic vascular resistance and watersodium retention caused by inhibition of renal prostaglandins.

A large and extensive meta-analysis conducted by an international group (CNT meta-analysis) analyzed data from randomized clinical trials of at least 4-weeks duration comparing NSAID with a placebo or another NSAID.33 Individual data, when available from the sponsors, was used. Data from 639 clinical studies were aggregated in an analysis of 68,342 person-years for placebo-comparisons and 165,456 person-years for comparisons between NSAIDs. Individualized data were available for 192,881 patients, mean-age  $61.2 \pm 11.3$  years, 63% treated for OA. The coxib group included celecoxib, rofecoxib, lumiracoxib, etoricoxib, and GW403681; this group was compared with diclofenac 150 mg daily (rarely 100 mg), ibuprofen 2400 mg per day, and naproxen 1000 mg (rarely 440 mg). The relative risk (RR) of hospital admission for HF in patients using high doses of NSAIDs compared with placebo was around two, regardless of the NSAID used (naproxen, diclofenac, ibuprofen, coxib).

A meta-analysis of observational studies demonstrated a significantly higher risk of HF in NSAID users [pooled risk ratio (RR)=1.17 (95% CI: 1.01-1.36)]. This increased risk was observed for traditional NSAIDs [RR=1.35 (95% CI: 1.15-1.57)], but there was no statistically significant risk of HF with coxibs (RR=1.03, 95% CI: 0.92-1.16). For the authors, this non-significant result could be related to insufficient sample size.<sup>34</sup>

The SOS project is a nested case-control study using five population-based healthcare databases from four European countries [the Netherlands, Italy, Germany, and the United Kingdom (UK)].<sup>35</sup> In this cohort of several million individuals with an average age of 77 years [standard deviation (SD): 11], the incident rate of HF was 37.5 per 10,000 person-years, comparable with that reported in population-based studies. Current NSAID users (within the previous 14 days) had an increased risk of hospitalization for HF (adjusted OR 1.19; 95% CI 1.17-1.22) compared with past use of any NSAID. This risk is increased by diclofenac, ibuprofen, indomethacin, ketorolac, naproxen, nimesulide, piroxicam, etoricoxib, and rofecoxib, but not by standard doses of celecoxib. This study indicates that this increasing risk of HF is dose-dependent for most NSAIDs.

HF is common in the very elderly. Its clinical symptoms may be very mild in patients with a sedentary lifestyle with reduced physical activity due to osteoarticular pathologies. Treatment with NSAIDs may result in decompensation of preexisting HF, leading to hospitalization. This exacerbation may be worsened by the interaction of NSAIDs with treatments (diuretics, angiotensin converting enzyme inhibitors, etc.) that reduce their effectiveness.

Atrial fibrillation (AF) is facilitated by age, high blood pressure, history of myocardial infarction (MI), or HF (which it aggravates), and can lead to thromboembolic complications such as strokes. Several studies have suggested that the use of NSAIDs increases the risk of AF. A populationbased prospective cohort study of elderly people [mean age 68.5 years (SD 8.7)] in the Rotterdam Study found that the current use of NSAIDs was associated with increased risk compared with never use (HR 1.76, 95% CI 1.07–2.88). Recent use (within 30 days after discontinuation of NSAIDs) was associated with an increased risk of AF compared with never use (HR 1.84, 95% CI 1.34-2.51).<sup>36</sup>

Stroke. In their meta-analysis of individual participant data from randomized trials for the CNT collaboration, there is no evidence that any NSAID significantly increases the risk of stroke.<sup>33</sup> A meta-analysis of observational studies published in 2015 concluded that, overall, patients on NSAID therapy did not have a significant risk of stroke.<sup>37</sup> However, there was a significant increase in risk among diclofenac or meloxicam users (RR 1.27; 95% CI, 1.02-1.59 and RR 1.27; 95% CI, 1.08-1.50, respectively). A recent populationbased case-control study, not included in this meta-analysis, also concluded that there was no increased risk of stroke with traditional NSAIDs as a group (OR=1.03; 95%CI, 0.90-1.19).<sup>38</sup> A significant increased risk was found with diclofenac and aceclofenac. This risk was greater at high doses, in the case of prolonged exposure and in patients with CV risks. The European case-control study in the SOS Project concluded that the use of coxibs (OR = 1.08, 95% CI 1.02-1.15) and use of traditional NSAIDs (OR = 1.16, 1.12–1.19) were associated with an increased risk of ischemic stroke.39 Significantly increased risks (in decreasing order) were found for ketorolac, diclofenac, indomethacin, rofecoxib, ibuprofen, nimesulide, diclofenac with misoprostol, and piroxicam. Naproxen, meloxicam, celecoxib, and ketoprofen showed no elevated risk. Ischemic stroke risk associated with NSAIDs seemed higher in vounger individuals, males, and those with a prior history of ischemic stroke. Surprisingly, the increase in risk would be most pronounced during the first 7 days of treatment, and, to a lesser extent, for treatments longer than 90 days.

*MI* and coronary risks. In the meta-analysis of the CNT collaboration, the risk of infarction or death from coronary heart disease was increased over placebo when a coxib was used (RR=1.76 95%CI, 1.31-2.37), diclofenac [adjusted risk ratio (aRR) = 1.70 95% CI, 1.19-2.41] or ibuprofen (aRR=2.22 95% CI, 1.10-4.48), but not naproxen (aRR=0.84 95% CI, 0.52-1.35).<sup>32</sup>

A Bayesian meta-analysis of individual patient data using Canadian and European healthcare databases found an increased risk of MI within the first 7 days of treatment, with no further increase in risk with longer treatment (except perhaps for diclofenac). There was a relationship between increasing NSAID daily dose and risk of acute MI. The ORs for the five NSAIDs studied were 1.24 (0.91-1.82) for celecoxib, 1.48 (1.00-2.26) for ibuprofen, 1.50 (1.06-2.04) for diclofenac, 1.53 (1.07-2.33) for naproxen, and 1.58 (1.07–2.17) for rofecoxib.<sup>40</sup> Concomitant use of cardioprotective aspirin does not appear to reduce the risk of acute MI with NSAIDs.41 A subsequent analysis using data from the Quebec administrative health cohort confirmed that all five NSAIDs, including naproxen, are associated with increased risk of MI. This increased risk exists for celecoxib after continuous use for more than 30 days, whereas, for the other four NSAIDs, it occurs within 7 days of use.42

The risk of acute MI was also studied in the SOS project. The study cohort included 8,535,952 new NSAID users, of whom 101,227 patients experienced an acute MI (AMI) after inclusion. Association with AMI was evaluated for 28 individual NSAIDs. The risk of MI with current use of any NSAID compared with past use of any NSAID is increased with OR=1.08 (95% CI 1.06-1.11). Estimates of the RR of AMI differ slightly between 28 individual NSAIDs. The RR was statistically significant for 12 of the 28 NSAIDs studied. It was highest for ketorolac, followed, in descending order of point estimate, by indomethacin, etoricoxib, rofecoxib, diclofenac, the fixed combination of diclofenac with misoprostol, piroxicam, and ibuprofen. It was correlated with COX-2 potency, but not only with coxibs. Higher doses showed higher risk estimates than lower doses; for duration, no clear patterns were seen.43

*Gl adverse events.* GI adverse events range from heartburn, dyspepsia, and bloating to major GI averse events, such as bleeding, ulceration, and perforation of the stomach or the intestine that can occur without any signal symptom and can be fatal.

Approximately half of patients who regularly take NSAIDs have gastric erosions, and 10–30% have gastric ulcers on endoscopy, but clinically significant problems are uncommon.<sup>44</sup> There is no correlation between NSAID gastropathy and upper abdominal symptoms frequently experienced by patients taking NSAIDs. The risk of GI complications increases with age and with history of peptic

ulcer disease or gastrointestinal hemorrhage, high doses of NSAIDs, and concomitant use of corticosteroids.<sup>44</sup>

The 2007 OARSI International COX-2 Study Group Workshop reported that several metaanalyses showed a RR of serious adverse GI effects in people who were current users of NS NSAIDs to be three-to-four times that of nonusers in the general population.<sup>45</sup>

A nested case control study conducted between January 2000 and 2005 found an adjusted RR of upper GI complications associated with current use of 3.7 (95% CI: 3.1–4.3) for traditional NSAIDs (tNSAIDs) and 2.6 (95% CI: 1.9–3.6) for coxibs. Daily dose was a predictor of increased risk for both tNSAIDs and coxibs. Users of tNSAIDs with a prolonged plasma half-life or slow-release formulations had an augmented risk of upper GI tract complications.<sup>46</sup>

A systematic review of observational studies on NSAIDs and upper GI bleeding/perforation published between 2000 and 2008 calculated a pooled RR estimates of upper GI bleeding/perforation of 4.50 [95% CI 3.82-5.31] for traditional NSAIDs and 1.88 (95% CI 0.96-3.71) for coxibs. The RR of upper GI tract hemorrhage/perforation varies among NSAIDs at doses commonly used in the general population. The RR was 2.69 (95% CI 2.17-3.33) for ibuprofen, 2.12 (95% CI 1.59–2.84) for rofecoxib, 1.44 (95% CI 0.65–3.2) for aceclofenac, 1.42 (95% CI 0.85-2.37) for celecoxib, 3.98 (95% CI 3.36–4.72) for diclofenac, 4.15 (95% CI 2.59-6.64) for meloxicam, 5.40 (95% CI 4.16-7.00) for indomethacin, 5.57 (95% CI 3.94–7.87) for ketoprofen, 5.63 (95% CI 3.83-8.28) for naproxen, 9.94 (95% CI 5.99-16.50) for piroxicam, and 14.54 (95% CI 5.87-36.04) for ketorolac. There was no significant correlation between this risk and the degree of COX-1 inhibition. Drugs with a profound and coincident inhibition (>80%) of both COX isozymes or with a long half-life or slow-release formulation were associated with a higher risk.<sup>47</sup>

A meta-analysis of nine randomized clinical trials comparing celecoxib with nonselective NSAIDs revealed that celecoxib had fewer GI side effects. However, this improved safety cannot be extrapolated to the entire coxib class,<sup>48</sup> and concomitant administration of celecoxib with low-dose cardioprotective aspirin often appeared to negate the GI-sparing advantages of celecoxib over NSAIDs. Also, in elderly patients ( $\geq$ 65 years, mean age: 71.9 years duration of treatment:  $\geq$ 2 weeks) digestive toxicity appears to be lower with celecoxib and diclofenac than with naproxen and ibuprofen.<sup>49</sup>

For the CNT meta-analysis, all NSAIDs increased upper GI complications (COX-2 inhibitors RR=1.81 95% CI 1.17–2.81; diclofenac RR= 1.89 95% CI 1.16–3.09; ibuprofen RR=3.97 95% CI 2.22–7.10; and naproxen RR=4.22 95% CI 2.71–6.56).<sup>33</sup>

There is no statistically significant difference in major GI side effects between nonselective NSAIDs and coxibs prescribed with proton pump inhibitors (PPIs).<sup>50</sup> PPIs used to protect the upper GI tract do not have a protective effect on the lower GI tract and may worsen NSAID-induced enteropathy.<sup>51,52</sup> This is of importance because it is estimated that more than two out of three patients taking long-term selective and nonselective COX-2 inhibitors develop enteropathy with intestinal inflammation, increased intestinal permeability, protein leakage, erosions, diverticula and risk of intestinal perforation, risk of occlusion, and diaphragm-like stricture.53 For some authors, these complications of the lower GI tract may be as frequent and severe as those of the upper tract.54

A national Spanish study, published in 2005, found a mortality rate of 5.6% in hospitalized patients due to serious events related to the upper or lower GI tract.<sup>55</sup>

Renal adverse events. Renal adverse events are important to consider in the very old. NSAIDs can induce hyperkaliemia, and sodium and fluid retention, especially in subjects with pre-existing diabetes mellitus, cardiac, renal, or hepatic impairment. This may result in peripheral edema, increased blood pressure and lower efficacy of antihypertensive treatments, congestive heart failure, and acute kidney injury (AKI). AKI triggered by NSAIDs is increased by concomitant use of a diuretic and/or a reninangiotensin inhibitor, and also in the dehydrated elderly with impaired renal function.56 In a recent meta-analysis, the OR of AKI due to current NSAIDs exposure in older people (>50 years) was 2.01 (95% CI 1.52–2.68) but lower: 1.73 (95% CI 1.32-2.29) with COX-2 inhibitors, although this difference was not statistically significant.<sup>57</sup> The risk of chronic renal failure associated with the use of NSAIDs was assessed by a systematic review and meta-analysis of observational general practice or population studies featuring patients aged 45 years and over. Regular-dose NSAID use did not significantly affect the risk of accelerated chronic kidney disease (CKD) progression; pooled OR=0.96 (95% CI: 0.86–1.07), but high-dose NSAID use significantly increased the risk of accelerated CKD progression; pooled OR=1.26 (95% CI: 1.06–1.50).<sup>58</sup>

### Risk of all-cause mortality in the elderly

Published studies do not provide a definite conclusion on the risk of mortality from any cause related to NSAID use. Thus, two studies, conducted using data from Australian veterans' administrative databases, reached different conclusions. For one, there is a reduction in all-cause mortality if an NSAID, whether selective or nonselective, has been used in the previous 2 years. The risk of death appears to be inversely related to the number of prescriptions provided.<sup>59</sup> For the other, there was a significantly higher mortality risk. The risk ratios and 95% CI relative to the control group (adjusted for age, sex, and CV risk) were: celecoxib 1.39 (1.25, 1.55), diclofenac 1.44 (1.28, 1.62), meloxicam 1.49 (1.25, 1.78), rofecoxib 1.58 (1.39, 1.79), nonselective NSAIDs 1.76 (1.59, 1.94).<sup>60</sup> The authors of the latter study indicate that these different results may be related, in part, to a more restrictive definition of NSAID exposure in their study than in the previous one.

In the meta-analysis of the CNT collaboration, using data from clinical trials conducted in a population with an average age of 61 years, the risk of death from any cause was significantly higher only in patients allocated to coxib, without excess risk for naproxen or ibuprofen, or even diclofenac, despite a clear increase in the risk of death from vascular causes for the latter.<sup>33</sup>

NSAIDs should certainly be discouraged in patients with severe renal impairment or heart failure. These are common conditions in the very old and should be routinely investigated before prescribing. NSAIDs should also be avoided in patients with a history of perforation, bleeding or GI ulcer, active peptic ulcer, hemorrhage, severe hepatocellular failure, coronary artery bypass graft, or history of aspirin or NSAIDs-related allergy or asthma.

Beyond these statements, what are current international recommendations from academic societies regarding the use of NSAIDs for OA management in the elderly?

## What do international OA recommendations for the very old say?

International recommendations for OA treatment make little or even no mention of age.<sup>61</sup>

The European League Against Rheumatism (EULAR) recommendations for knee OA support paracetamol as the first line, and, if successful, the preferred long-term oral analgesic.<sup>1</sup> NSAIDs should be considered in patients unresponsive to paracetamol. In the case of increased GI risk, nonselective NSAIDs and gastroprotective agents or coxibs should be used.

The American College of Rheumatology (ACR) recommends "considering topical NSAIDs prior to use of oral NSAIDs".<sup>2</sup> However, ACR strongly recommends oral NSAIDs in knee, hip, and hand OA, without any mention of age.

The OsteoArthritis Research Society International (OARSI) updated recommendations do not recommend oral NSAIDs in knee OA in the context of frailty (but they do not mention age). NSAIDs with more favorable safety profiles may be used in frail patients in hip and polyarticular OA, at the lowest possible dose for the shortest possible duration.<sup>3</sup>

In the updated European Society for Clinical and Economic Aspects of Osteoporosis and Osteoarthritis (ESCEO) algorithm for the management of knee OA,<sup>4</sup> in the case of insufficient efficacy of a first step treatment (including SYSADOA: symptomatic slow acting drugs for osteoarthritis that are supposed to reduce the need for analgesics), ESCEO takes age into account and recommends, for considerations of safety, favoring topical over oral NSAIDs, particularly in patients aged 75 years old and over.

The American Geriatric Society (AGS) does not specifically target OA but focuses on persistent pain management in older patients with a strong recommendation<sup>5</sup>: traditional or selective NSAID use for chronic pain, should be rare, considered in highly selected individuals, and only if the foreseeable risks appear to be outweighed by therapeutic benefits. Peptic ulcer, chronic renal failure, or HF are absolute contraindications. In patients with hypertension, *Helicobacter pylori* infection, a history of peptic ulcer, concomitant use of steroids or inhibiting serotonin-reuptake antidepressants, NSAIDs should be avoided. Indomethacin should never be used in elderly patients.

Baseline age (years)	Major vascular events			Symptomatic upper GI events		
	Allocated coxib	Allocated placebo	RR (99% CI)	Allocated coxib	Allocated placebo	RR (99% CI)
<60	50/10327	24/6866	1.44 (0.75–2.77)	44/10332	8/6873	2.74 (1.22–6.12)
60-69	84/7862	38/5632	1.51 (0.91–2.51)	54/7879	14/5647	2.11 (1.04–4.28)
70–79	103/6565	65/6522	1.42 (0.93–2.16)	42/6598	27/6542	1.35 (0.68–2.68)
≥80	45/1198	27/1191	1.49 (0.77–2.88)	20/1194	8/1200	2.31 (0.80-6.70)

**Table 1.** Number of CV and symptomatic upper GI adverse events by age group, for coxib and placebo. Additional unpublished data provided by the coxib and traditional NSAIDs collaboration with permission.

CI, confidence interval; CV, cardiovascular; GI, gastrointestinal; NSAIDs, nonsteroidal anti-inflammatory drugs; RR, relative risk.

In 2012, the AGS updated the list of drugs whose potential risks outweigh the benefits in the elderly.<sup>6</sup> NSAIDs are stigmatized for their high GI risk in patients over 75 years or if they are taken concomitantly with corticosteroids, anticoagulants, or antiplatelet agents.

The Society specifies that this list is not intended to be used in a punitive manner. In any case, clinical judgement is irreplaceable and mandatory to allow for individualized adapted and shared therapeutic decisions between the therapist and the patient.

All these recommendations do not preclude using NSAIDs in very old OA patients without contraindications and severe co-morbidities. Although therapeutic indications appear to be limited, prescribing NSAIDs is possible for patients suffering from pain in whom first-line treatments have failed to significantly reduce pain and when predictable risks appear to be overweighed by therapeutic benefits.

Their prescription requires caution, careful monitoring, and, above all, the consent of the patient after complete information on the risk-benefit ratio. However, this assumes knowing the accurate quantified estimates of the GI, CV, and renal relative and absolute risks.

## Do we have data on NSAIDs RRs in the elderly?

A review of observational studies conducted at the request of United States (US) health officials concluded that the RR of serious CV or GI side effect was independent of age.<sup>62</sup> Age undoubtedly increases the absolute risk but NSAIDs use-RR remains unchanged whatever the age. The authors of the CNT meta-analysis came to the same conclusion: the RR of serious secondary events when using NSAIDs is not related to age.32 We asked them to provide these rates according to age categorization: <60, 60-69, 70-79, $\geq$ 80 years.<sup>63</sup> In their answer,<sup>64</sup> they confirmed they observed no evidence of a greater RR of serious CV event for COX-2 inhibitors (p=0.88), naproxen (p=0.62), or other NSAIDs (p=0.18), and no evidence of a greater RR of serious symptomatic upper GI side effect p=0.28, p=0.37, and p=0.30, respectively, in any of these age groups (Table 1). The conclusions of this metaanalysis of trials in selected patients are consistent with those of observational studies: NSAIDs increase CV and GI risks by a multiplying factor that can be estimated, and varies upon the risk and the NSAID, but is not dependent on age or other baseline patient characteristics.

## Do we know what NSAIDs absolute risks are in the very old?

For care providers, the basic incidence of serious CV and GI diseases in the very old not using NSAIDs is difficult to ascertain. National or private insurance systems have databases. As far as we know, they have not yet reported accurate data on this major issue.

An estimate of the absolute incidences can be drawn from the results of placebo groups from meta-analyses, but since patients with high CV or GI risk are excluded from trials their results cannot be extended to the real population.

Several scores to predict CV risk have been developed. The Heart Score of the European Society of Cardiology and the Framingham Coronary Heart Disease Risk Score are not applicable beyond a certain age. The CV Risk Calculator of the American College of Cardiology and the American Heart Association is applicable to African-American and non-Hispanic white individuals between 40 and 79 years.<sup>65</sup> For older people, it only estimates a 10-year risk for atherosclerotic CV disease.

The QRISK, developed by Hippisley-Cox and based on the English/Welsh population, seems the only valid score in patients aged 30–84 years old. It allows for estimating the CV risk in the coming year.<sup>66</sup>

For GI pathologies, a British population-based controlled analysis, also developed by Hippisley-Cox, provides incidence of peptic ulcer or GI hemorrhage by sex and age in the general population.<sup>67</sup>

The combination of the age-related basic risk with the RR related to NSAID use provides an estimate of the incidence of adverse events and deaths due to NSAIDs (Tables 2 and 3). We are fully aware that this risk calculation remains imprecise, particularly due to the lack of epidemiological data on the incidence of CV and GI events in the general population. However, this data provides an order of magnitude that can be useful to the patient in making an informed decision about whether or not to take an NSAID and whether or not to consider this risk acceptable for symptom relief.

### Conclusion

We believe there is room for NSAIDs in the treatment of OA in the very old. The disease is painful, disabling, and severe enough to justify, in some cases, the risks. **Table 2.** Incidence of major vascular and severe upper GI events, expressed per 100 patient–years, in the placebo group of the CNT meta-analysis,<sup>32</sup> and according to HC epidemiological studies,<sup>65,66</sup> for a male subject in the UK, with no history but with treated hypertension.

	Age (years)	Placebo group	НС			
Major vascular	70–79	1.0	2.0-3.3			
Events	≥80	2.3	3.0-4.0			
Severe upper GI	70–79	0.4	0.2-0.4			
Events	≥80	0.7	0.4-0.8			
GI, gastrointestinal; HC, Hippisley-Cox; UK, United Kingdom.						

A small percentage of very old OA patients will respond to NSAIDs and experience a clinically significant improvement. They are the ones who will buy NSAIDs OTC or ask for a prescription from their healthcare provider. These patients should be fully aware of the risks and warnings regarding NSAID treatment. Recent data suggests the possibility of providing patients with quantitative information on CV and GI risks related to age, co-morbidities, or other risk factors and related to NSAID use. This may allow for making a shared decision, taking into account the patient's preferences and personal beliefs.

New evidence suggests that the frequency and severity of side effects may vary across drugs used. The prescriber, therefore, has to choose the treatment best suited to each patient's profile.

The prescription of NSAIDs in very old OA patients should respect their contra-indications and always be as short as possible, at the lowest effective dose, and under strict clinical and biological monitoring.

**Table 3.** Based on data from CNT meta-analysis and HC epidemiological studies<sup>32,65,66</sup>: RR of AEs during NSAIDs therapy and incidence per 100 patient-years of AEs observed, NSAIDs-related AEs, and NSAID-related death.

	Age (years)	Placebo group <sup>18</sup>	NSAID RR	Incidence AEs	Incidence NSAID- related AEs	NSAID-related death
Major vascular	70-79	1.0	0-1.4	1-1.4	0-0.4	0-0.1
Events	≥80	2.3	0-1.4	2.3-3.2	0-0.9	0-0.3
Severe upper GI	70-79	0.4	2-4	0.8-1.6	0.4-1.2	0.008-0.024
Events	≥80	0.7	2-4	1.4-2.8	0.7-2.1	0.014-0.042

AEs, adverse events; GI, gastrointestinal; HC, Hippisley-Cox; NSAIDs, nonsteroidal anti-inflammatory drugs; RR, relative risk.

#### Research agenda

Some questions still need to be addressed, amongst which:

- The need to access epidemiologic data on the frequency of the risk of GI, major CV events and AKI in the general population, according to age categorization, from public institutions, national health services, and private insurance databases.
- The need, while the published results remain contradictory, to better specify for each NSAID (and not only for the entire class of traditional NSAIDs or Coxibs) the relative frequency of the different risks.

#### Acknowledgements

The French AGRHUM group (Association Geriatric and RHeUMatology) are Christian Cadet, Emmanuel Maheu, Philippe Bréville, Pierre Emmanuel Cailleaux, Claude Jeandel, Jean-Laurent Le Quintrec, Agnès Portier, Odile Raynaud-Levy, Bernard Verlhac, and Jean-Marie Vetel, who all contributed to this work.

### **Conflict of interest statement**

The authors declare that there is no conflict of interest.

### Funding

The authors received no financial support for the research, authorship, and/or publication of this article.

#### **Ethical issues**

Our review did not require ethics committee approval because it was based on available, previously published information and analysis of datasets obtained from other researchers whose data were appropriately anonymized and whose informed consent had been obtained at the time of initial data collection.

### ORCID iD

Christian Cadet D https://orcid.org/0000-0002-6625-2011

#### References

 Jordan KM, Arden NK, Doherty M, et al. EULAR recommendations 2003: an evidencebased approach to the management of knee osteoarthritis: report of a task force of the Standing Committee for International Clinical Studies including Therapeutic Trials (ESCISIT). *Ann Rheum Dis* 2003; 62: 1145–1155.

- 2. Kolasinski SL, Neogi T, Hochberg MC, *et al.* 2019 American College of Rheumatology/ Arthritis Foundation guideline for the management of osteoarthritis of the hand, hip, and knee. *Arthritis Care Res (Hoboken)* 2020; 72: 149–162.
- 3. Bannuru RR, Osani MC, Vaysbrot EE, et al. OARSI guidelines for the non-surgical management of knee, hip, and polyarticular osteoarthritis. *Osteoarthritis Cartilage* 2019; 27: 1578–1589.
- Bruyère O, Honvo G, Veronese N, et al. An updated algorithm recommendation for the management of knee osteoarthritis from the European Society for Clinical and Economic Aspects of Osteoporosis, Osteoarthritis and Musculoskeletal Diseases (ESCEO). Semin Arthritis Rheum 2019; 49: 337–350.
- American Geriatrics Society Panel on Pharmacological Management of Persistent Pain in Older Persons. Pharmacological management of persistent pain in older persons. *J Am Geriatr* Soc 2009; 57: 1331–1346.
- American Geriatrics Society 2015 Beers Criteria Update Expert Panel. American Geriatrics Society 2015 updated beers criteria for potentially inappropriate medication use in older adults. *J Am Geriatr Soc* 2015; 63: 2227–2246.
- Le Quintrec JL, Verlhac B, Cadet C, *et al.* Physical exercise and weight loss for hip and knee osteoarthritis in very old patients: a systematic review of the literature. *Open Rheumatol J* 2014; 8: 89–95.
- Tubach F, Ravaud P, Baron G, et al. Evaluation of clinically relevant states in patient reported outcomes in knee and hip osteoarthritis: the patient acceptable symptom state. Ann Rheum Dis 2004; 64: 34–37.
- Da Costa BR, Reichenbach S, Keller N, et al. Effectiveness of non-steroidal anti-inflammatory drugs for the treatment of pain in knee and hip osteoarthritis: a network meta-analysis. *Lancet* 2017; 390: e21–e33.
- Bannuru RR, Schmid CH, Kent DM, et al. Comparative effectiveness of pharmacologic interventions for knee osteoarthritis: a systematic review and network meta-analysis. Ann Intern Med 2015; 162: 46–54.
- Moore RA, Derry S, Simon LS, et al. Nonsteroidal anti-inflammatory drugs, gastroprotection, and benefit-risk. Pain Pract 2014; 14: 378–395.

- 12. Osani MC, Vaysbrot EE, Zhou M, *et al.* Duration of symptom relief and early trajectory of adverse events for oral nonsteroidal antiinflammatory drugs in knee osteoarthritis: a systematic review and meta-analysis. *Arthritis Care Res (Hoboken)* 2020; 72: 641–651.
- Solomon DH, Rassen JA, Glynn RJ, et al. The comparative safety of analgesics in older adults with arthritis. Arch Intern Med 2010; 170: 1968–1978.
- Avouac J, Gossec L and Dougados M. Efficacy and safety of opioids for osteoarthritis: a meta-analysis of randomized controlled trials. *Osteoarthritis Cartilage* 2007; 15: 957–965.
- 15. Kroon F, Kortekaas M, Boonen A, *et al.* Results of a 6-week treatment with 10 mg prednisolone in patients with hand osteoarthritis (HOPE): a double-blind, randomised, placebo-controlled trial. *Lancet* 2019; 394: 1993–2001.
- Wenham CY, Hensor EM, Grainger AJ, et al. A randomized, double-blind, placebo-controlled trial of low-dose oral prednisolone for treating painful hand osteoarthritis. *Rheumatology* (Oxford) 2012; 51: 2286–2294.
- Marcum ZA, Perera S, Donohue JM, et al. Analgesic use for knee and hip osteoarthritis in community-dwelling elders. *Pain Med* 2011; 12: 1628–1636.
- Nuesch E, Dieppe P, Reichenbach S, *et al.* All cause and disease specific mortality in patients with knee or hip osteoarthritis: population based cohort study. *BMJ* 2011; 342: d1165.
- 19. Hawker GA, Croxford R, Bierman AS, *et al.* Allcause mortality and serious cardiovascular events in people with hip and knee osteoarthritis: a population based cohort study. *PLoS One* 2014; 9: e91286.
- Corsi M, Alvarez C, Callahan LF, et al. Contributions of symptomatic osteoarthritis and physical function to incident cardiovascular disease. BMC Musculoskeletal Disord 2018; 19: 393.
- 21. Castell MV, van der Pas S, Otero A, *et al.* Osteoarthritis and frailty in elderly individuals across six European countries: results from the European Project on OSteoArthritis (EPOSA). *BMC Musculoskeletal Disord* 2015; 16: 359.
- Richardson CG, Chalmers A, Llewellyn-Thomas HA, *et al.* Pain relief in osteoarthritis: patients' willingness to risk medicationinduced gastrointestinal, cardiovascular, and cerebrovascular complications. *J Rheumatol* 2007; 34: 1569–1575.
- 23. Häuser W and Tölle TR. Meta-analyses of pain studies: what we have learned. *Best Pract Res Clin Rheumatol* 2015; 291: 131–146.

- Warner TD and Mitchell JA. COX-2 selectivity alone does not define the cardiovascular risks associated with non-steroidal anti-inflammatory drugs. *Lancet* 2008; 371: 270–273.
- Sánchez Delgado EJ. Life without COX 2 inhibitors: risks and benefits are determined by dose and potency. *BMJ* 2006; 332: 1451–1452.
- Moore N. Coronary risks associated with diclofenac and other NSAIDs: an update. *Drug Saf* 2020; 43: 301–318.
- Horwitz RI and Feinstein AR. The problem of "protopathic bias" in case-control studies. *Am J Med* 1980; 68: 255–258.
- Frishman WH. Effects of nonsteroidal antiinflammatory drug therapy on blood pressure and peripheral edema. *Am J Cardiol* 2002; 89: 18D–25D.
- 29. Johnson AG. NSAIDs and blood pressure. Clinical importance for older patients. *Drugs Aging* 1998; 12: 17–27.
- Pope JE, Anderson JJ and Felson DT. A metaanalysis of the effects of nonsteroidal antiinflammatory drugs on blood pressure. *Arch Intern Med* 1993; 153: 477–484.
- Pavlicevi I, Kuzmani M, Rumboldt M, et al. Interaction between antihypertensives and NSAIDs in primary care: a controlled trial. Can J Clin Pharmacol 2008; 15: e372–e382.
- 32. Girerd X, Hanon O, Pannier B, et al. Hypertension artérielle du sujet âgé en France: caractéristiques du traitement et fréquence de la plainte cognitive selon l'enquête FLAHS 2014. Ann Cardiol Angéiologie 2015; 64: 145–149 (in French).
- 33. Bhala N, Emberson J, Merhi A, et al. Vascular and upper gastrointestinal effects of non-steroidal anti-inflammatory drugs: meta-analyses of individual participant data from randomised trials. Lancet 2013; 382: 769–779.
- 34. Ungprasert P, Srivali N and Thongprayoon C. Nonsteroidal anti-inflammatory drugs and risk of incident heart failure: a systematic review and meta-analysis of observational studies. *Clin Cardiol* 2016; 39: 111–118.
- 35. Arfe A, Scotti L, Varas-Lorenzo C, et al. Nonsteroidal anti-inflammatory drugs and risk of heart failure in four European countries: nested case-control study. BMJ 2016; 354: i4857.
- Krijthe BP, Heeringa J, Hofman A, et al. Nonsteroidal anti-inflammatory drugs and the risk of atrial fibrillation: a population-based follow-up study. BMJ Open 2014; 4: e004059.

- Ungprasert P, Matteson EL and Thongprayoon C. Nonaspirin non- steroidal anti-inflammatory drugs and risk of hemorrhagic stroke: a systematic review and meta-analysis of observational studies. *Stroke* 2016; 47: 356–364.
- García-Poza P, de Abajo FJ, Gil MJ, et al. Risk of ischemic stroke associated with non-steroidal anti-inflammatory drugs and paracetamol: a population-based case-control study. J Thromb Haemost 2015; 13: 708–718.
- 39. Schink T, Kollhorst B, Varas Lorenzo C, et al. Risk of ischemic stroke and the use of individual non-steroidal anti-inflammatory drugs: a multicountry European database study within the SOS project. PLoS One 2018; 13: e0203362.
- Bally M, Dendukuri N, Rich B, *et al.* Risk of acute myocardial infarction with NSAIDs in real world use: Bayesian meta-analysis of individual patient data. *BMJ* 2017; 357: j1909.
- 41. Bally M, Nadeau L and Brophy JM. Studying additive interaction in a healthcare database: case study of NSAIDs, cardiovascular profiles, and acute myocardial infarction. *PLoS One* 2018; 13: e0201884.
- 42. Bally M, Beauchamp ME, Abrahamowicz M, et al. Risk of acute myocardial infarction with real-world NSAIDs depends on dose and timing of exposure. *Pharmacoepidemiol Drug Saf* 2018; 27: 69–77.
- Masclee GMC, Straatman H, Arfè A, et al. Risk of acute myocardial infarction during use of individual NSAIDs: a nested case-control study from the SOS project. *PLoS One* 2018; 13: e0204746.
- Laine L. Nonsteroidal anti-inflammatory drug gastropathy. *Gastrointest Endosc Clin N Am* 1996; 6: 489–504.
- Moskowitz RW, Abramson SB, Berenbaum F, et al. Coxibs and NSAIDs-is the air any clearer? Perspectives from the OARSI/International COX-2 Study Group Workshop 2007. Osteoarthritis Cartilage 2007; 15: 849–856.
- 46. García Rodríguez LA and Barreales Tolosa L. Risk of upper gastrointestinal complications among users of traditional NSAIDs and COXIBs in the general population. *Gastroenterology* 2007; 132: 498–506.
- Massó González EL, Patrignani P, Tacconelli S, et al. Variability among nonsteroidal antiinflammatory drugs in risk of upper gastrointestinal bleeding. *Arthritis Rheum* 2010; 62: 1592–1601.
- 48. McCormack PL. Celecoxib: a review of its use for symptomatic relief in the treatment of

osteoarthritis, rheumatoid arthritis and ankylosing spondylitis. *Drugs* 2011; 71: 2457–2489.

- Mallen SR, Essex MN and Zhang R. Gastrointestinal tolerability of NSAIDs in elderly patients: a pooled analysis of 21 randomized clinical trials with celecoxib and nonselective NSAIDs. *Curr Med Res Opin* 2011; 27: 1359–1366.
- 50. Wang X, Tian HJ, Yang HK, et al. Meta-analysis: cyclooxygenase-2 inhibitors are no better than nonselective nonsteroidal anti-inflammatory drugs with proton pump inhibitors in regard to gastrointestinal adverse events in osteoarthritis and rheumatoid arthritis. Eur J Gastroenterol Hepatol 2011; 23: 876–880.
- Wallace JL. Mechanisms, prevention and clinical implications of nonsteroidal anti-inflammatory drug-enteropathy. *World J Gastroenterol* 2013; 19: 1861–1876.
- 52. Gwee KA, Goh V, Lima G, *et al.* Coprescribing proton-pump inhibitors with nonsteroidal anti-inflammatory drugs: risks versus benefits. *J Pain Res* 2018; 11: 361–374.
- Lanas A, Panés J and Piqué JM. Clinical implications of COX-1 and/or COX-2 inhibition for the distal gastrointestinal tract. *Curr Pharm Des* 2003; 9: 2253–2266.
- Utzeri E and Usai P. Role of non-steroidal antiinflammatory drugs on intestinal permeability and nonalcoholic fatty liver disease. World J Gastroenterol 2017; 23: 3954–3963.
- 55. Lanas A, Perez-Aisa MA, Feu F, *et al.* A nationwide study of mortality associated with hospital admission due to severe gastrointestinal events and those associated with nonsteroidal antiinflammatory drug use. *Am J Gastroenterol* 2005; 100: 1685–1693.
- 56. Dreischulte T, Morales DR, Bell S, et al. Combined use of nonsteroidal anti-inflammatory drugs with diuretics and/or renin-angiotensin system inhibitors in the community increases the risk of acute kidney injury. *Kidney Int* 2015; 88: 396–403.
- 57. Zhang X, Donnan PT, Bell S, *et al.* Nonsteroidal anti-inflammatory drug induced acute kidney injury in the community dwelling general population and people with chronic kidney disease: systematic review and meta-analysis. *BMC Nephrol* 2017; 18: 256.
- Nderitu P, Doos L, Jones PW, et al. Nonsteroidal anti-inflammatory drugs and chronic kidney disease progression: a systematic review. Fam Pract 2013; 30: 247–255.

- 59. Mangoni AA, Woodman RJ, Gaganis P, et al. Use of non-steroidal anti-inflammatory drugs and risk of incident myocardial infarction and heart failure, and all-cause mortality in the Australian veteran community. Br J Clin Pharmacol 2010; 69: 689–700.
- Kerr SJ, Rowett DS, Sayer GP, et al. All-cause mortality of elderly Australian veterans using COX-2 selective or non-selective NSAIDs: a longitudinal study. Br J Clin Pharmacol 2011; 71: 936–942.
- 61. Maheu E, Monod P and Bard H. Les nouvelles recommandations 2014 de l'OARSI pour le traitement non chirurgical de la gonarthrose. Lecture et point de vue critiques. *Les Cahiers de l'année gérontologique* 2015; 7: 74–77 (in French).
- 62. Chou R, McDonagh MS, Nakamoto E, et al. Analgesics for osteoarthritis: an update of the 2006 comparative effectiveness review. Rockville, MD: Agency for Healthcare Research and Quality (US), 2011.

- 63. Cadet C, Maheu E and on behalf of the French AGRHUM Group. Coxibs and traditional NSAIDs for pain relief. *Lancet* 2014; 383: 121–122.
- Bhala N, Emberson J, Patrono C, et al.; on behalf of the CNT Collaborators Group. Coxibs and traditional NSAIDs for pain relief. *Lancet* 2014; 383: 121–122.
- 65. ACC/AHA ASCVD Risk Calculator; http://www. cvriskcalculator.com
- 66. Hippisley-Cox J, Coupland C, Robson J, Brindle P. Derivation, validation, and evaluation of a new QRISK model to estimate lifetime risk of cardiovascular disease: cohort study using QResearch database. *BMJ* 2010; 341: c6624.
- Hippisley-Cox J, Coupland C and Logan R. Risk of adverse gastrointestinal outcomes in patients taking cyclo-oxygenase-2 inhibitors or conventional non-steroidal anti-inflammatory drugs: population based nested case-control analysis. *BMJ* 2005; 331: 1310–1316.

Visit SAGE journals online journals.sagepub.com/ home/tab

**SAGE** journals